Author’s response to reviews

Title: Ventilatory and ECMO treatment of H1N1-induced severe respiratory failure: results of an Italian referral ECMO center.

Authors:

Giovanni Cianchi (giovannicianchi@yahoo.it)
Manuela Bonizzoli (manbon@libero.it)
Andrea Pasquini (pasquinia@aou-careggi.toscana.it)
Massimo Bonacchi (massimo.bonacchi@unifi.it)
Giovanni Zagli (giovanni.zagli@unifi.it)
Marco Ciapetti (ciapettim@aou-careggi.toscana.it)
Guido Sani (guido.sani@unifi.it)
Stefano Batacchi (sbatac@tin.it)
Simona Biondi (simona.biondi@yahoo.it)
Pasquale Bernardo (bernardop@aou-careggi.toscana.it)
Chiara Lazzeri (chiara.lazzeri@unifi.it)
Valtere Giovannini (giovanniniv@aou-careggi.toscana.it)
Gian Franco Gensini (gensinigf@unifi.it)
Adriano Peris (aperis@libero.it)

Version: 7 Date: 25 November 2010

Author’s response to reviews:

Re: MS: 1608630004020891

Dear Editor,

This revised manuscript has been amended according to the points raised by the Referee #3. Language has been edited by an English native speaker. A point by point reply to the Referee’s comments is enclosed.

We hope that our paper can be now of interest for your Journal.

Yours sincerely,

Giovanni Zagli (corresponding author of behalf of all the authors)

Answer to Referee #3

We thank Referee #3 for his/her constructive criticisms and comments to our manuscript.

The use of lung ultrasound has been now emphasized (modified parts in blue for an easier verification), as suggested. However, other areas were not modified, according with the requests and opinions of the other 3 Referees.
Regarding the evaluation of the threshold for access to the ECMO procedure, we can say that the timing decision to begin treatment for H1N1-induced ARDS cases was always based on the National Protocol of the Ministry of Health, as stated in the manuscript.